Electric vehicles based electric power grid support: a review

Afaf Rabie, Abdelhady Ghanem, Sahar S. Kaddah, Magdi M. El-Saadawi

Department Electrical Engineering, Faculty of Engineering, Mansoura University, Mansoura, Egypt

Article Info

ABSTRACT

Article history:

Received Sep 10, 2022 Revised Nov 27, 2022 Accepted Dec 11, 2022

Keywords:

EV Frequency control Grid support Power quality Voltage control Grid connected electric vehicles (EVs) can provide energy quality services to ease intermittent renewable energy sources dependent, enhance grid stability performance, reliability and load balancing in the power system. Moreover, incorporating sufficient EVs into the power grid will assist reduce greenhouse gases. However, the unrestrained EVs charging impact the raise of peak request, frequency deflection, voltage instability, power quality from the acceptable limits. It can also cause overloading of the power system equipment and an increase in power losses. In this paper, the influences EV technologies on various energy systems, control scheme strategy, benefits, and motivational challenges are investigated. An overview of the impacts of uncontrolled EV loading on the electric power system and how controlled charging becomes benefit impacts is presented. Moreover, a review of the controlled charge and discharge benefits of EVs and the electrical serving involved in reduction of frequency deviation, voltage stability, power quality improvement is introduced. This review shows that integration of optimal control of EVs with a suitable optimization technique can improve and provide on time multi ancillary services of EV integrated with weak grid. In addition, this review provides research tracks which can be followed by researchers.

This is an open access article under the <u>CC BY-SA</u> license.



Corresponding Author:

Abdelhady Ghanem Department Electrical Engineering, Faculty of Engineering, Mansoura University Aldaqahlyia Mansoura, 35516 Egypt Email: aghanem_m@mans.edu.eg

1. INTRODUCTION

Transportation using fossil fuels accounts for around 26% of global CO₂ emissions, and this percentage is projected to rise rapidly as the industry grows [1], [2]. Nowadays, different countries' environmental pollution management strategies encourage a variety of transportation-related sources to improve energy quality and reduce emissions [3]. Interest in electric vehicles (Evs) has spread in the current era as another potential for vehicles powered by Gasoline, with the tender of various sorts of EVs. Due to environmental issues, governments drive EV encouragement, and hence, EVs are expected to dominate the passenger car market during incoming years [4]. Currently, a large part of oil consumption has been allocated in the transportation sector, where fuel-powered vehicles use a large proportion of oil according to the International Energy prospect. In the transportation sector, oil consumption will increase by 54% until 2035 [1], [5]–[7]. The total number of EVs is expected to reach 130 million by 2030, according to the International Energy Agency's "Global Prospects for EVs" report [8]. The intended transformation in the fundamental energy source from gasoline to the electricity grid has created numerous challenges for power grids around the world. For instance, by 2050, the National Grid (NG) in [9].expects the electricity demand for EVs to reach about 600 billion kilometers annually in Britsh streets. If it is assumed 75% of these

forecasted distances will be provided by electric powered vehicles of 0.15 kWh/km, A 54 TWh is required which represent an increase of 18% as compared to the UK present consumption.

Most of the EV technologies were widely used and gained a lot of observation in the past years. In the last century, lead-acid batteries were used to derive fuel-powered vehicles combined with internal combustion engine cars (ICEVs). However, in recent years so many improvements are accomplished on EVs and batteries to improve their performance and reduce their cost [7]. Several terms are used for EVs based on their multiple levels of electrification. A hybrid EV (HEV) improves overall efficiency by combining a normal car with an EV propulsion system (combustion engine propulsion system) [2], [10]. While in the plug-in EV (PEV) conventional internal combustion engine vehicles is converted to HEVs. A plug-in hybrid vehicle (PHEV) connects to the electric grid to recharge its battery and then uses the battery as the source of motion. In this case, the normal car engine can provide back source propulsion whenever the charged battery is exhausted. A battery EV (BEV) relies on the power grid and does not have an internal combustion engine and receives power from the onboard battery [11]. On the other hand, the on-board electric motor of a Fuel Cell EV (FCEV) is powered by a fuel cell, depending on a small battery or a super capacitor. The electricity in the FCEV is generated by oxygen from compressed air. EV classified by their alternative fuels and technologies are presented in [2], [11]-[13]. In this paper, the acronym "EV" is used to refer to any type of vehicle that can be integrated into the network in order to recharge the battery or for saving coincident services to the network.

EV can help solve environmental problems by reducing greenhouse gas emissions and noise pollution [14], [12]. In addition, an EV with recharging batteries is a dispersed supply of energy, however, its integration with the network presents both benefits and challenges. From the customer's point of view, the small EV range and the shortage of charging stations are the biggest challenges [15]. For this reason, fast-charging stations will be placed where the cars could have long EV parks and the range will also be expanded [16]. The need and the progress required in the growth of EVs were discussed in [2], [4] and they indicated that EV has a vivid future and highlighted the significance of considering the challenges related to this transformation. The power system will be influenced directly by this transformation, Therefore the EV integration and its impacts in the context of a power system must be considered [16].

Several researchers have evaluated the economic cost and economic benefits of EVs over utility and distribution systems [10]. A global review of practices related to EV consumer education activities to explore actions that governments can take to promote market growth and to better understand how to implement such campaigns were assessed in [17], [18]. They summarized practices in consumer consciousness, and how to transact with EV. On the other hand, many studies demonstrated that even low penetration of EV charging can affect power system quality which includes voltage instability, frequency drift, losses, system thermal load, loss of transformer life, demand response, and harmonic distortion [19].

Many initiatives have been taken so far to mitigate these quality problems. Different charge control strategies were presented for improving system stability. Jadhav and Kalkhambkar [20], designed an intelligent control strategy to reduce the maximum load and improve the load profile based on the quadratic programming of EV was presented.

Several authors have analyzed the effect of EVs on the distribution network and the EV technologies concept was presented in detail in [21]. In the literature, most of the research work was focused on smart charging methodology, commercial models, and load profiles. This paper presents the concept of EV technologies, and various topics that can be explored in EV field technology. The main objective of this study is to examine the effects of EVs on the frequency regulation and voltage stability of the distribution network. The organization of the rest of this paper is as follows: Section 2 introduces aspects related to the integration of EVs into the power system and their impact on the energy quality of the system. Section 3 investigates the impacts of EVs on voltage stability in electric power systems and how to employ them to improve power quality. Section 4 deals with system frequency regulation depending on EVs. Section 5 discusses salient aspects and the trend research. Finally, Section 6 concludes the paper.

2. EV INTEGRATION IN POWER SYSTEM

In all EVs, the main source is the battery and characterized by weight, size, and cost component. DC batteries are electrochemical power sources that receive, store, and deliver direct current electrical energy [22]. Although it is important to understand the basics of chemistry, from an EV designer's perspective, the battery contains a set of performance-based criteria. Some of these criteria are battery state of charge, charging efficiency, charge/discharge depth of battery, and lifespan of battery.

Another important aspect of EVs is the charging station and its charging schemes. EV charging framework is categorized according to 'the types of charging' and 'the modes of charging'. Types are indicated by plugs and sockets that support power transmission. Modes refer to fast or slow charging modes

where the consumption of power source to EV varies. There are four types of charging modes in which the EV substructure is categorized. Those mode types mainly depend on the power source, communication availability, and protection of the battery [23], [24].

The vehicle is plugged into ordinary house power or industry power sockets in charging mode 1 and 2, providing a current of no more than 13-16 A. A 230 V AC power supply from the distribution feeders to the house is suitable for charging the EV. Residual current device (RCD) protection that prevents fatal shocks is not available in mode 1. However, (RCD) protection is provided in the cable prepared by the manufacturer of EV in mode 2. In charging mode 3, there is an intended charging terminal with an install of the residual current device (RCD) and stream protection connected to the mains. This type of mode supports a single-phase AC power 16 A or 32 A. Compared to other modes, this mode is the preferred charging one due to its protection, scalable power features, and connectivity [4], [25]. In addition, the EV is rapidly charged using DC current in charging mode 4. Due to the high output voltage and current of 500 V and 200 A, 80% of the EV battery can be charged within 30 minutes. The charging stations installation costs of this mode are high and they cause overload of the distribution network [11].

It is worth mentioning that allowing EVs to charge/discharge without any control technology causes grid disturbances, voltage, and frequency fluctuations. On the other hand, when EVs are intelligently charged/discharged, they can support the quality of the electric power grid. In this direction, it is possible to take benefits from connecting EVs with the electric networks to improve the performance of the networks, which is called vehicle to grid (V2G) technology [15]. V2G technology allows EV batteries to store energy and discharge it back to the grid when it is most needed (i.e. peak times). V2G is a technology that consists of transmitting energy in two ways, either from the vehicle to the network or from the network to the vehicle [26]. This technology can help to regulate charging and discharging between EVs (EVs) and electric networks in order to increase the voltage stability of the distribution networks [27].

Figure 1 shows the general block diagram of the V2G architecture [27]. Due to the special characteristics of the EV, the grid operators around the world countenance significant challenges in integrating the EV load that differ from other loads in the grid [26]. It is not just about charging an EV from the grid compared to home charging, but it also has specific dynamic behavior. Furthermore, the real and reactive power that EV will use cannot simply be assessed, due to the difference in time and place where the EV will be connected to charge, and the duration of charging required [15]. EV is connected to the power system as a charging load or an EV battery works as a potential fellow used as a power source that creates a V2G technology.



Figure 1. Block diagram of the V2G architecture [27]

According to the abovementioned discussions, it is important to pay attention to EV concerns with the power grid. The following substations present a general review of existing EV network impact studies. Several studies have discussed concerning policies and processes used to reduce the influence associated with EVs battery charging/discharging on a large scale [28]. A comprehensive literature review is also presented about research and developments concerning vehicles' charging/discharging and their integrations with the electric distribution networks. The conclusion of this part detects the gaps and durability of the current studies and makes a proposal for potential next work [13].

2.1. Impacts of EV on the distribution power system elements

Nowadays a large number of EV charging loads are connected to distribution networks. The EV load characteristics indicate that there will be an increased tension on elements of the weak network system [26], [28]. This issue is reviewed in this section to gain a clear vision of the probable and expected

troubles of EV charging and discharging on an electric distributed system. Figure 2 shows a setup of an integrated power grid for the onboard charging scheme of EV with type 1 and 2 AC charging schemes [29]. Although EV gives a clear advantage to transportation, its impact on power systems cannot be condoned. Unregulated EV charging influences the distribution system with reference to its voltage profile, grid unbalance, power loss, frequency stability, as well as harmonic disturbance. Many research studies have identified the problems by suggesting several ways to control the charging of EVs [30]. Al-Ogaili *et al.* [31] briefly reviewed EV control charging strategies. They categorized EV charging control strategies into three approaches of clustering, scheduling, and forecasting. The models of EV charging control conditions are highlighted to evaluate and compare the methods used in charging EV. Moreover, the merits and demerits of the different applied methods were identified. However, dealing with a discrepancy between frequency and voltage drop is an important issue and it required more attention to EVs that respond together to a frequency and voltage signal. Lopes *et al.* [32] used a supervisory control scheme for local voltage and frequency based on EVs was introduced. The control points were sent to each vehicle or set of vehicles from the master console. The paper also introduced the power group grading to control the frequency and voltage drop.



Figure 2. A facility integrated power grid setup for a type1 and type 2 AC charging scheme for an EV [5]

2.2. Positive impacts of EVs on electric power systems

This section reviews the effects of EV on power quality. A grid analysis on the district power grid was performed in Ada County, USA, for a projected 18% load stimulated to be equivalent to available EVs by 2040 [33]. EVs that are parked for days are linked to the charging ports for more than the standard recharge time. So, a battery of EV can be utilized to reserve network grid services and earn income for EV users by the supplied energy into the grid to maintain the demand or by the guided charging time and power consumption to reduce electric bill and cost of EV charging [7]. The smart grid voltage is regulated depending on EVs in [34] and an algorithm tis proposed to maximize the parking lot revenue in the line distribution network taking into account the charging demand of EV owners and voltage fluctuations, where the voltage is determined by the power distribution flow method [34]. Several studies have shown that controlled charging of EVs can enhance the efficiency of the power system [35], minimize operating costs [36], and reduce the contractility of renewable energy systems (RESs) [37]. The research in [38] aims to studies the system stability and EVs energy consumption by applying a modern control techniques. The controller of artificial neural network is proposed to estimate the required yaw direct torque to stabilize the EV dynamics with four internal motors for wheels. Furthermore, discharge control of EVs can improve electrical services and provide additional benefits. EVs can provide network servings due to the rapid reply of chargers like primary frequency control (PFC) and secondary frequency control (SFC) and reduced power loss due to the high capacity of the battery [39]. An overview of the positive and negative impacts of EVs as well as EV modeling as a grid load are presented in [40].

2.3. Power system issues with high penetration of EV charging in distribution networks

Although the ecological and economic advantages of EVs are evident, their effects on power systems quality still need to be analyzed [41]. System power quality is an important key in the security and

efficiency of networks, particularly weak networks, and expect to be pretentious by the improvement of EVs over the upcoming years. EV interface systems use electrical transducers because of their infrastructure and the nature of the transducer power electronic components, which are highly nonlinear systems [41], [42]. Therefore, there are usually several dissimilar ways for reducing and dealing with the negative effects of the EV on the weak network. PEV can be used in power quality improvement, in both charge and discharge modes, generate good power quality (in terms of voltage optimization and frequency regulation).

The power quality of the system elaborates the potentiality of the system to keep synchronizing when tolerating any disturbances [18]. Understanding of the difficulties needs studying of the power quality issues. Change in generation and load, negative interaction among controllers, and fast exciters are reasons for system vacillations [18], [19].

Deflections must be mitigated, as the disturbances to which they have been subjected cause excessive wear and tear on the components of the power system. Moreover, these disturbances can increase in proportion, and if uncontrolled they may lead to unreliability and then power outages [43].

A comprehensive study with low and high penetration of EVs was conducted in a public low-voltage (LV) distribution network. The studies in [35], [44], [45] investigated the location, duration, and time of charge for EVs. An overload of distribution transformers was found once a day in winter [44]. This overload only took place in the summer with a high penetration level of EV load.

Vatandoust *et al.* [45] found that due to a Nissan leaf charger (6.6 KW) with 30% penetration, the system parameters can absorb the charging at type 1 of charging (120 V). But when charging at type 2 (240 V), there would be a big daily variation in the amplitude of heat on the cable which reduced the cable lifetime significantly. Another investigation in a local network [10] found that an increased level of EV penetration leads to an increase in overloaded lines.

The conclusion of the study system in [46] indicated that by 2025, its distribution system could reliably meet about 500,000 HEV penetrations. Likewise, a study of a system in Australia (Perth) [47] proved that existent high voltage (HV) system infrastructure had a significant potential to facing the entire permeation of EV market. On the other hand, in [24], [44] authors discussed the merit and demerit effects of charging EV on the lifetime of the distribution transformer. On one hand, they found an increase in mean operating temperature which led to insulating failures of transformers. furthermore, the coordinated load profile acquired by controlling the EVs charging lowered the temperature of the transformers. This was due to the reduced numeral of extensions and limitations of the transformer in a day.

In contrast, [28] has identified EV charging as a way to use the distribution feature in an environment of mobile smart grid. The capacity to scale up for stimulated energy demand due to EVs may be limited by the weakness of the grid. They comprise the capacitance and thermal constraints of the network segments. The study in [28] concluded that the voltage drop registered in some regions fell less than the standard constraints, and therefore, the system needed to be boosted. Subsequently, the studies must identify such limitations in each relevant network component.

Comparably, the presented study in [29] found that there is a voltage distract of 12.70-43.30% due to the estimated EV penetration of 20.0% and 80.0% for various rates of charging. Dulau and Bica [15] introduced an experimental model of a distribution system with connected EVs was presented and revealed that an on load tap commutators (OLTC) model could withstand a 20% penetration of EV load, but the voltage dropped lower than the limitation constraints in various locations at a 30% penetration of EV. However, the model voltage disturbance was kept within constraints except for the case of charging with a high current. A comprehensive study presented in [13], proved that in transmission and distribution systems there were obvious impacts on voltage even at low levels of penetration as low as 1-2% due to EVs located randomly in the grid. In another study on a low voltage borders were transcended at the network end until a minimum load charge occurred in the network [48].

A study of a network with IEEE 34-bus at 24.9 kV presented in [49], confirmed that voltages on buses could be kept within limits of up to 8% of EV penetration for different charging scenarios. Researchers in [42], [50] have looked at the uncontrolled V2G mode of powering EVs. They found that voltage limits violated the permissible limits during minimum grid load terms. The search in [42] showed that OLTC can bear penetration of 10% but cannot tolerate penetration of 30%. In a medium-voltage system in a network with a voltage of 15 kV presented in [51], a similar result was obtained. As the level of penetration of electric chargers increased, there were voltage deviations in the overhead buses, as expected. Moreover, it revealed that although the considered system cannot withstand uncontrolled charging with a penetration of 10%, it can accommodate a double tariff structure with an EV penetration level of 13% [51], [52].

A controlled algorithm was proposed to regulate the network voltage, depending on the compensation of reactive power from EV [53], [54]. Decentralized coordination of EV charging and EV load management has been proposed in [55] to overcome negative influences on voltage stability. Finally, it was

clear that voltage regulators may be useful based on the charging and network characteristics to achieve voltage stabilization requirements as explained in [46], [56].

3. FREQUENCY STABILITY USING INDEPENDENT EV

The frequency of the power system is an important indicator of the balance of active power supply and request that must be maintained at nominal values for ordinary operation. Power imbalance takes place because of steady load versions or fluctuations in the RES era relying on atmospheric situations at the ordinary operation [40]. But power imbalance arises in emergencies when loads, generating units, or transmission lines are suddenly interrupted. Network shocks caused by big-scale loading and load shedding have a substantial impact on the electrical grid quality, power losses, and equipment usage [20], [40].

In traditional networks, weak grids are common in medium or low voltage levels. Weak networks feature with low inertia and high resistance compared with strong networks, also rated as exemplary low power classifications [57]. EVs can regulate the frequency of the distribution system as well as participate in power change of peak load. Frequency regulation may be realized locally via the transfer of active energy between the distribution network and the EV battery, which is controlled by a two-way power electronic converter [58]. The power electronics converter of EVs can be controlled in order to serve as frequency primary control.

In the primary frequency control (PFC) there are two control loops, droop control that reduced the dependence on traditional generators, which helps in reducing fossil consumption, and control the inertial that simulates the behavior of a traditional generator [37]. Figure 3 shows EV control block diagram for frequency regulation [39], [59]. Frequency is regulated in a conventional power system using synchronous generators in big stations. But in future power systems, adjustable loads such as EVs and heat pumps play an important role in frequency management [37]. The power network's inertia can compensate for load disruptions or imbalances induced by intermittent renewable sources. However, keeping frequency variations within acceptable ranges remains a challenge [60].



Figure 3. V control block diagram for frequency regulation [39]

Under such a transient state of affairs, it is critical to decrease frequency variations and restore the quality to the network's regular state as quickly as possible [61]. Many frequency regulation solutions examine the V2G approach to manage the frequency of a micro grid (MG)/weak grid with various levels of EV penetration [37]. A proposed methodology was presented in [62] to reduce the effect of EV on a 'weak' grid via frequency regulation is explored based on the control system of EV. The metering system based on the smart algorithm shown in Figure 4 is integrated along with the EV control scheme. Additionally, frequency regulation (SFR) was participated with many EVs through control of V2G approach considering both the control center organization and the state of charge (SOC) levels of battery anticipated from EV users in [63].

In addition, EV has evident advantages in the frequency regulation of the power system from a financial point of view for reducing the EVs charging/discharging costs [36], [64], [65]. However, controlling frequency in distribution feeders depended on a bi-directional V2G converter with two-stage for EV, SOC is disregarded, as well as battery age and health are ignored [58]. Additionally, EVs can be used for a one-way power flow network in frequency regulation service [66]. Given the power flow rate and the charging demands of EV owners, a decentralized V2G control approach was selected to include EVs in frequency management. Based on the best-integrated grid, the SOC battery control method hold effectively the EV battery SOC, whilst EV was concerned with regulating system frequency in the grid [59].



Figure 4. Flow chart of the proposed algorithm [62]

The distributed control strategy is discussed in [67], [68] to meet the expectations of battery SOC from EV by taking the charge state control approach of the EV battery to conserve the remaining battery SOC. The frequency deviation of the system is affected by the co-distribution, as shown in [69]. Ota *et al.* [69] determined the charging energy depending on the genetic frequency change data.

Ota *et al.* [67], used half of the maximum V2G power was distributed ignoring the difference between the particular and scheduled duration of the plug-in. The demand for EV charging was considered, and V2G-based decentralized frequency regulation was developed based on EV. Moreover, Due to the quick response of electronic interface of the EV, chargers of EV batteries respond faster than conventional generating units. So, the charging and discharging of controlled EVs can be an effective choice for regulating the grid frequency [25].

An experimental test was carried out on the potential of commercially available EVs (e.g., Nissan Leaf) to offer primary frequency control (PFC) only by altering charging power and without V2G abilities [70] The results demonstrated the technical ability of EVs to achieve a PFC with a fast response time when a small, insulated, renewable power system was used as a test system. When EVs are shared in the load, the frequency deviations on both sides of generation and load respond by AGC. The potential and appropriate operation of AGC with EVs was investigated to substitute the characteristic response inability of conventional generating units [71]. Almeida *et al.* [71] have established two methods of engaging EVs in the PFC. Turning off EV charging was the first and the second method was to inject energy into the network in V2G mode. In addition, the V2G central control method for co-load frequency control (LFC) has been discussed. Tian *et al.* [72], a method was designed to track the LFC signal. However, there was no thought given to explain how to send the LFC signal to EVs. So an area control model (ACE) was instituted and connected two-area models containing V2G that were used to resolve the impacts of participating EV in secondary frequency regulation depended on ACE. The results showed the effectiveness of V2G for reducing the frequency variation and tie-line flow fluctuation by lowering the reserve capacity in the system.

A coordinated LFC control strategy and battery SOC has been proposed in [37], [73] based on EVs operating in V2G mode with conventional generation. The suggested control technique was put to the test on the British electricity grid for frequency regulation by coordinated LFC control between EV controller and conventional power plants (PPs). The priority in response was for EV controller than the controller of PPs when the system exceeded the permissible limits. Also, a proposed stability standard was improved to

evaluate the stability for the LFC system while the inertia unsureness and delays in time-varying are considered simultaneously. In addition, the LFC of EV used a PI controller for frequency stability enhancement. The findings demonstrated the efficacy of the suggested method in enhancing frequency control and reducing energy mismatch. EVs operation based on two charging control modes was studied in [74] to regulate frequency taking into account the diving behavior of the EV owner. The first mode simply regulated charging and the second regulated both charging and discharging of EVs. Results revealed that EVs were successful in lowering frequency oscillations. The dynamic frequency regulation frame of EV is indicated in Figure 5 for the two modes of the strategy [74].



Figure 5. Dynamic EV frequency regulation frame with two control modes [74]

Wang and Chen [75] proposed a dual-level assented-based frequency control procedure to support PFC. The first control level aimed to reduce multi-zone power system frequency fluctuation, and the second one aimed to decrease the cost of regulating frequency and degradation expense of battery for individual EVs. The efficiency of EVs was tested to provide basic frequency regulation for an isolated power system with diesel generators, wind turbines, and hydro generators [76]. The study demonstrated the effectiveness of EVs in regulating frequency oscillation while controlling the energy consumed by EVs and coordinating charging time. Moreover, the effects of probably delayed actions were estimated. Donadee and Ilic [77], proposed a dynamic stochastic programming method was used to improve the charging decision of EVs and frequency regulation, considering the unpredictability of EV and wind power.

Furthermore, the output fluctuations of the traditional power were decreased as a result of EV integration in the system as an energy storage system. The EVs can engage in LFC in the isolated micro grid that operated in an autonomous mode of operation [78]. While economic distribution is a fundamental issue in operation and management, EVs are involved in frequency control based on economical transmission [78]. The economic dispatch model was created based on Incentive policy to match EVs and the grid operator [79]. The results showed that an additional network of EVs with an appropriate dispatch plan succeeded to close the gap between demand and supply at different time intervals and reduce fuel costs. While contracts and tariffs were modified to implement V2G integration, the implications of various forms of contracts on various marketplaces were examined [80]. A driver incentive scheme was designed, as well as "whole-deal" business models for potential EV participants. Long-term contracts provided a greater motivation than short-term committing. The predicted capabilities were ideal for the regulatory nodes between the network operator and the EV aggregator to create a queueing model for EVs and design a smart mechanism for charging [80].

Smart pricing policies and mechanisms are able to improve the performance of frequency stability. On the other hand, some academics have suggested methods to stimulate the performance of EVs in frequency regulation based on the control cost mechanism [81]. It was able to recharge many EVs and adopted the non-linear pricing of the model. Some researchers have developed a price system to incentivize cars to voluntarily connect V2G on their own. Moreover, the mechanisms effectively set the proportional

importance between the control of SOC and the revenue [81]. A control structure of real-time was suggested to perform the V2G system into the price-based schedule to manage the system effectively [82]. The collector profits from the contracted capacity of the sale and allocated the advantages to each EV participating in the program by developing an incentive scheme to coordinate benefits between the network side and the consumer side at frequency regulation [82].

In conclusion, EVs can support frequency regulation by controlling vehicle charging without V2G abilities or controlling the charging and discharging based on V2G capacity. The initial technique achieved frequency control solely by changing the power of EVs charging that required simple infrastructure and had little impact on battery life. In the second technique, frequency management is achieved by regulating both charging and discharging energy. This technique is more sophisticated, necessitates infrastructure upgrades such as bidirectional chargers, and has an impact on the battery life cycle owing to constant charging and discharging. After investigating the research that clarifies the impact of the EVs on frequency regulation, the next section represents the effect of the EVs on the voltage support.

4. IMPACTS OF EVS ON VOLTAGE STABILITY IN ELECTRIC POWER SYSTEMS

Even though voltage stability is a key problem in power system operations and planning due to the rising frequency or voltage disturbance occurrences, voltage stability associated with EV load integration is a very critical challenge in power system operation. The load parameters are the most important variables influencing voltage stability as shown in many searches [30], [82], [83]. However, few researches have studied the effect of EV loading on voltage stability in the system, considering the load of EV as a constant current load [84]. The work in [83] examined EV behavior by treating the load of EV as continuous power demand and analyzed experimentally whether coordinated integration of EVs could reduce voltage disturbance by using a smart charging algorithm relying on a drooping controller. The results showed the benefits of smart EV charging in enhancing the voltage stability of an extremely unstable network. The effect of V2G performance on network voltage stability was estimated in [44]. A single-phase EV charger was modeled to study the effect of EV chargers in a LV distribution feeder and a voltage analysis was estimated in [44]. The impacts on the feeder with various EV penetration levels as well as different load levels were studied. Through the analysis of results, the efficiency of the EV charger was improved by suggesting a suitable candidate eliminate the issues and include them in future work.

Since the voltage stability of a power system is greatly affected by the characteristics of the load, it is necessary to represent an accurate load as introduced in [48]. The voltage response of four various kinds of EVs was estimated and models of ZIP were improved for each and then they were utilized in V2G studies. The evaluations proved that there was a significant difference in behavior between the EVs types and thus a generic model cannot be accurate [48].

Therefore, a loading model for a fast charger of EVs was developed based on a common battery charger at the first stage, the voltage stability was examined with the improved model [85]. It has been proven that the system voltage limits were significantly influenced by the EV model. The second stage of that study illustrated how EV charging affected the stability of system voltage considering limits of voltage stability, power losses, and the flow ratings for a suitable place of EV fast-charging stations in the network. Moreover, the maximum permissible capacity of the station was estimated depending on the stability and reliability of the system. Due to numerous economic and environmental considerations, existing power systems are operating at or near their stability limitations [86]. Ultra-fast charging (UFC) technology has attracted great interest in research and industry. Thus, planning the locations and capacities of the UFC stational charging stations. Next, it discussed the expectations for UFC planning outcomes by identifying patterns similar to the renewable energy (RES) source pattern. The pattern considered the stochastic properties and energy density of RES and UFC. The study focused on UFC design from a grid perspective, traffic flow, and EV owner behavior for potential integration of UFC inside smart cities [86].

EV load has characteristics that combine variable and constant exponential power load behaviors [23]. During system voltage drops due to system disturbance, passive exponential loads consume more power, which significantly degrades system voltage stability. Hence, estimating the influence of EVs charging on the power supply voltage stability is critical. Therefore, including voltage stabilization into the EV charging infrastructure development process is advantageous from a grid standpoint [9], [87].

4.1. Mitigating impacts of EV on voltage stability in the system

For a specific power grid, voltage stability is acceptable if all buses voltage are remained in their constraints during ordinary operation and after exposure to disturbances [88]. Significant voltage instability occurs when the system generation and transmission are not sufficient to meet system demand, and this results in the overloading of transmission lines. While the consumption of reactive power increases due to

increased loads or due to a decrease in reactive power generation, crises in the system occur [84]. Allowing EVs charging or discharging without any control technology causes voltage deviation and system disturbances, but when EVs are smartly charged/discharged, they achieved to support the power system [3]. Many studies have been concentrated on how to utilize V2G technology to help stabilize the system voltage and thus how to mitigate the EV influence on stability. Figure 6 shows a block diagram of power quality management in the grid depending on EV technology [83]. Mohamed *et al.* [89] introduced a study was performed to determine the most effective location of a weak bus and a strong bus, to assign a vehicle station within the specific grid based on the controller output that has been modified, considering bus voltage and EV state of charge (SOC). A fuzzy voltage-dependent controller (FLC) was designed to notice its effect on the power system regarding voltage stability when the controller was just dependent on the charging status of the EV's batteries or bus voltage [89].



Figure 6. A block diagram of power quality management in the grid depending on EV technology [83]

Researches [42]–[44], [50], [90], [91] studied the methods used to take advantage of coordinated charge control of EVs to reduce their impact on system power quality. Increased system peak requirements, voltage stability limit violations, increased power system losses, and potential overloading of transformers, cables, and lines are among the identified network effects. An available EV charger was designed in Matlab to study EV chargers in and an inclusive harmonic analysis was performed in a low voltage distribution feeder. The feeder was influenced by different levels of EV penetration that charge at various load levels. From that study, it was shown that the system can bear up to 30% penetration of EV charger at normal load and up to 40% penetration of EV charger could be absorbed to the distribution feeder without exceeding system constraints. Karmaker *et al.* [43] the impact of EV charging stations on the distribution networks in Bangladesh was analyzed considering total power demand, voltage sag, voltage swell, harmonics, and transformer losses. The power quality disturbances were also analyzed to mitigate the negative influence of EV load. When EV chargers were integrated into the distribution system, the harmonic disturbances got higher. Thus, the THD of connecting one EV was about 4.82%, for three EVs about 12.35%, and for five EVs, and with various specifications, THD reached about 19.69% [43].

A coordinated charging was suggested in [44] to reduce power loss and increase the load factor. The optimum charging profile for PHEVs was calculated by reducing losses. Since it was not possible to accurately predict household loads, stochastic programming was presented in the study. quadratic and dynamic programming were the two main techniques that were analyzed. The power loss was rather negligible if the household load profiles were known. Results were obtained with the quadratic and dynamic programming techniques. The effect of EVs on a typical low-voltage network has been studied in [50]. A probabilistic approach was used to model the uncertainties associated with EVs behaviors and to assess charger location, times, and duration of charging. Two levels were used to penetrate EVs. Also, the study checked the usual days of the week in winter and summer for each level. Energy streams were run in half-hour time steps Sequentially for each of the four conditions. overheating of transformer, cable, voltage drops, and losses associated with each case were reported. Simulation results of a UK urban public grid model demonstrated that the transformer and ground cable serving 96 customers were significantly overburdened in terms of EV penetration levels. The high penetration level of EVs caused a voltage drop of less than 0.94 p.u. and increased energy loss [50].

A study of EVs charging effect on a low voltage (LV) distribution network in New Toshka city, Aswan, Egypt for different EV penetration levels (12.5%, 25%, 50%) was implemented in [90]. Two schemes of charging were investigated delayed and unsupervised charging. To conduct the study a power factor simulation software was utilized. The EV charging impact on loading transformer and cable, voltage stability at the end point in the system, and power loss were evaluated. The results showed that delayed charging has fewer effects on the low voltage network than uncontrolled charging.

A model of neighborhood electricity demand of ten houses was prepared in [91] depending on the home data, EV type, traveling miles, and time of leaving /arriving of EV. The study developed also a thermal model to determine the temperature of the hot spot and transformer life loss. The results showed that charging at type 1 (120 V) has little effect on transformer aging whereas charging at level 2 has a bad effect on the transformer as it failed because of overheating.

4.2. Voltage stabilization depending on EV control charging and discharging

Different components can be used to manage active energy including DGs, storage technologies, and programmable devices such as EVs and heat pumps. On the other hand, OLTC, capacitors, and static var compensation can be used to regulate reactive power. Several researchers have investigated the regulation of voltage stability using effective EV control based on conducting online tests of the regulated charging approach [92]. The goal was to increase the content of EV owners while keeping the distribution network restrictions in mind. Voltage drift, power loss, and transformer load were all reduced as a result of the method. Figure 7 shows a block diagram of a vehicle controller for voltage support [93].



Figure 7. Voltage-controlled EV charging scheduled [93]

Nour *et al.* [94], proposed a decentralized/autonomous charge-control mechanism has been proposed. It suggested that, if the voltage was normal, EV charged at a rapid rate, but if the voltage was low, charging should be lowered or stopped. The power of charging was controlled depending on local voltage, battery SoC, while the charging precedence was given to EVs with lower SoC. In comparison to unregulated charging, the suggested approach decreased voltage loss and improved voltage profile. Mara *et al.* [95], suggested an EV-controlled charging technology was used to solve the problem of integrative voltage. A strategy to mitigate voltage spikes in feeders with PV, depending on a coordinated EV load as a storage system has been introduced. Results demonstrated the effectiveness of utilizing a coordinated EV load in feeders with PV to reduce voltage instability problems.

Another method for dependent voltage regulation of EVs is the use of capacitors in the DC link of bidirectional EV chargers to deliver reactive power. Even if the EV is disconnected to charge, the EV charger can accomplish this as illustrated in [96]. The EV bidirectional charger capacity was evaluated at type 1 to offer reactive power assistance. Therefore, the V2G process was illustrated by simulating various modes of operation the most important of which was the reactive power supply with/without PHEV battery charging. According to the findings, a DC jumper capacitor could offer reactive power support without any influence on the battery power system.

The direct voltage control method used in [53] to investigate the effect of the voltage according to the DC rapid charging of an EV in the low voltage distribution network during the condition of peak load.

The simulation displayed that the fast charging of six EVs pushed the grid to exceed the voltage constraints. Therefore, a two-way DC fast-charging station with a new control algorithm was proposed to solve the problem of voltage drop, manage the voltage and decrease power loss.

A technique for regulating voltage in the distribution network was suggested in [97], which depended on the coordination between EVs, DGs format, and OLTC to provide reactive power support operating in the V2G scheme. The suggested method has proven effectiveness in regulating voltage, reducing OLTC operation periods, and reducing system active loss.

From the results of the above review, unregulated charging of EVs can cause large voltage dips in networks, in addition to violating voltage permissible limits, particularly in long-distance charging in feeders. It also necessitates upgrading the network's infrastructure such as the installation of voltage regulators. However, by employing regulated charging and discharging methods, the voltage in all areas of the distribution network may be kept within constraints, and the voltage profile can be improved without the need for voltage regulators.

5. DISCUSSION

Normally, low and medium voltage networks provide low inertia. Therefore, its weakness is considerable effect on the frequency and voltage drift/stability. EV is one important key to overcome fast variation as well as violation of the voltage and frequency limits via its designed battery control. Many researchers have investigated the operation of regulated and unregulated grid connected EV. Unregulated EV with high capacity introduces frequency as well as voltage fluctuation with weak grid during its plug in and plug off. Therefore, regulated EV solves this problem and provides seamless connection with weak grid and improves the voltage profile and frequency change when other heavy loads interconnected with the grid. Many aspects have been investigated such as frequency control, voltage control, power quality, economic impact, cost and optimal control as well as its integration with conventional operation and control of EV. Moreover, investigation of grid connected EV limit with specific grids over the world considering its power electronics interfacing components as well as passive components and its effect on the overall grid power quality (especially voltage and frequency control) have been reviewed. Table 1 (in Appendix) gives most related research/contribution during last years for grid connected EVs with ancillary services.

Yet, investigation of EV integration with multi-bus weak grids and its optimal location in order to improve the voltage profile, losses and voltage long term stability are hot research areas. In addition, more exploration based on low level control of EV in order to improve its integration with very weak grid as well as increase virtually system inertia are running. Moreover, application of multi-object optimization techniques for multi-job grid connected EV during its charging/discharging mode as well as multi agent systems, hourly electricity-based bill price, load profile and weather/generation modes are good research points.

6. CONCLUSION

EVs introduce many advantages to the grid (V2G mode) and generate revenue for EV users. However, operation of EV with power grid faces many challenges issues and are need to be properly addressed. The objective of this paper is to announce and address the latest researches and their results of activities related to different aspects of grid support depending on EV. This paper reviewed the current state of design of EVs, including EV types, batteries used in EV technology, and charging modes. The recent researches explain the function of the EV as a catalyst in improving the power grid support, as the EV represents a load on the network and has harmful effects and faces challenges. It also has positive and effective impacts in improving the power grid voltage and frequency. In addition, this work reviewed recent literature regarding the impact of EV charging stations on the power grid and EV role in improving its voltage and frequency profile. The negative effects of uncontrolled EV charging stations on the grid such as voltage instability and frequency deviation have been clarified. Finally, this paper reviewed and analyzed the works on the positive effect of EV charging stations that are achieved by implementing controlled V2G technology. It can be concluded that grid connected EV can be used to improve many aspects of electrical network based on proper control of its own battery. However, proper design and optimization techniques are required in order to ensure life and cost of the battery, proper plug in/off of EV without violation of grid limits and finally optimal location of its connection with the grid. Therefore, research on integration such EVs with different new technologies into weak grids based on multi object optimization and/or intelligent control will be conducted.

APPENDIX

Table 1. Most related research/contribution during last years for grid connected EVs with ancillary services		
Reference	Contribution	Scope
[37], [70],	The primary frequency control (PFC) is employed to reduce fossil consumption, and control	Frequency regulation
[71], [75],	system inertial	
[76]		
[60]	The power network's inertia compensated load disruptions or imbalances induced by	
	intermittent renewable sources while keeping frequency variations within acceptable ranges remains a challenge	
[62]	V2G control technique using proposed methodology for reducing the effect of EV on a 'weak' grid via frequency regulation is explored based on the control system of EV	
[63]	Supplementary Frequency Regulation (SFR) is utilized with V2G control approach considering both the control center organization and the SOC levels	
[36], [64],	Based on financial control aspects, EVs charging/discharging costs is proposed.	
[65], [78], [80]		
[58]	bi-directional V2G converter with two-stage for EV is used ignoring the SOC and battery age.	
[59], [67]–	decentralized V2G control approach for best-integrated grid considering the battery SOC	
[13] [72]	A control model (ACE) was instituted using DI controllar for frequency stability enhancement to	
[13], [72],	resolve the impacts of participating EV in secondary frequency regulation depended on ACE	
[75]	frequency control procedure to support PEC and decrease the cost of regulating frequency and	
[/3]	degradation expense of battery for individual FV	
[81] [82]	Control of SOC with the revenue are employed for incentivizing cars to voluntarily connect	
[01], [02]	V2G on their own	
[83]	Depending on a smart charging algorithm, EV impact is examined for reducing voltage instability	Effect of EV on power system quality
[48]	Voltage response of four various kinds of EVs is estimated	F
[86]	Ultra-fast charging (UFC) technology considering traffic flow, and EV owner behavior is studied due to numerous economic and anyiconmental considerations	
[0] [87]	voltage stabilization into the EV charging infrastructure is advantageous from a grid standpoint	
[50] [90]	Effect of EVs on a typical low-voltage network studied with high penetration level of EVs	
[91]	which proved that there was a voltage drop of less than 0.94 p. u. and increased energy loss	
[21]	A fuzzy voltage dependent controller (FLC) is evaluated to determine the most effective	Voltage support
[07]	location of a weak bus and strong bus	voltage support
[42]_[44]	Coordinated charge control of EVs to reduce their impact on system power quality with 30%	
[50] [90]	30% penetration of EV charger at normal load	
[91]	50% peretudion of ET charger at normal road	
[43]	Impact of EV charging stations on the distribution networks is studied considering total power	
F 4 41	demand, voltage sag, voltage swell, harmonics, and transformer losses	
[44]	A coordinated charging control is employed using quadratic and dynamic programming to	
[02]	reduce power loss and increase the load factor	
[92]	Voltage stability is regulated using effective EV control based on conducting online tests of the	
[02]	regulated charging approach	
[93]	voltage support Based on a vehicle controller keeping the distribution network restrictions in	
10.41	mind. Voltage drift, power loss, and transformer load	
[94]	A decentralized/autonomous charge-control mechanism is proposed with determining the	
	cnarging manner.	
[53] [95]	voltage instability problems are reduced depending on EV-controlled charging technology	
[96]	A capacitor in the DC link of bidirectional EV chargers is used to deliver reactive power support	
	without any influence on the battery power system	
[97]	voltage regulation with reducing OLTC operation periods, and reducing system active loss is achieved depending on the coordination between EVs, DGs format	

REFERENCES

- International Energy Agency, "Global EV Outlook 2021," *Global EV Outlook 2021*, p. 101, 2021, [Online]. Available: https://iea.blob.core.windows.net/assets/ed5f4484-f556-4110-8c5c-4ede8bcba637/GlobalEVOutlook2021.pdf. [1]
- [2] C. A. Tracker, "1.5 C-consistent benchmarks for enhancing Japan's 2030 climate target," 2021.
- M. Falahi, H. M. Chou, M. Ehsani, L. Xie, and K. L. Butler-Purry, "Potential power quality benefits of electric vehicles," IEEE [3] Transactions on Sustainable Energy, vol. 4, no. 4, pp. 1016–1023, 2013, doi: 10.1109/TSTE.2013.2263848.
- J. A. Sanguesa, V. Torres-Sanz, P. Garrido, F. J. Martinez, and J. M. Marquez-Barja, "A review on electric vehicles: Technologies [4] and challenges," Smart Cities, vol. 4, no. 1, pp. 372-404, 2021, doi: 10.3390/smartcities4010022.
- [5] F. Mwasilu, J. J. Justo, E. K. Kim, T. D. Do, and J. W. Jung, "Electric vehicles and smart grid interaction: A review on vehicle to grid and renewable energy sources integration," Renewable and Sustainable Energy Reviews, vol. 34, pp. 501-516, 2014, doi: 10.1016/j.rser.2014.03.031.
- W. A. Salah et al., "Electric vehicle technology impacts on energy," International Journal of Power Electronics and Drive [6] Systems, vol. 10, no. 1, pp. 1–9, 2019, doi: 10.11591/ijpeds.v10n1.pp1-9. G. R. C. Mouli, P. Venugopal, and P. Bauer, "Future of electric vehicle charging," *19th International Symposium on Power*
- [7] Electronics, Ee 2017, vol. 2017-Decem, pp. 1-7, 2017, doi: 10.1109/PEE.2017.8171657.

- [8] Z. Zhu and H. Du, "Forecasting the Number of Electric Vehicles: A Case of Beijing," IOP Conference Series: Earth and Environmental Science, vol. 170, no. 4, 2018, doi: 10.1088/1755-1315/170/4/042037.
- [9] A. Mohammad, R. Zamora, and T. T. Lie, "Integration of electric vehicles in the distribution network: A review of PV based electric vehicle modelling," *Energies*, vol. 13, no. 17, 2020, doi: 10.3390/en13174541.
- [10] J. J. A. Saldanha, E. M. dos Santos, A. P. C. de Mello, and D. P. Bernardon, "Control Strategies for Smart Charging and Discharging of Plug- In Electric Vehicles," *Smart Cities Technologies*, 2016, doi: 10.5772/65213.
- [11] Y. Kongjeen, W. Junlakan, K. Bhumkittipich, and N. Mithulananthan, "Estimation of the quick charging station for electric vehicles based on location and population density data," *International Journal of Intelligent Engineering and Systems*, vol. 11, no. 3, pp. 233–241, 2018, doi: 10.22266/IJIES2018.0630.25.
- [12] S. George, R. V. Chacho, and K. Salitha, "Modelling and simulation of Electric Vehicle power train in SEQUEL," 2014 IEEE International Conference on Power Electronics, Drives and Energy Systems, PEDES 2014, 2014, doi: 10.1109/PEDES.2014.7042149.
- [13] F. Un-Noor, S. Padmanaban, L. Mihet-Popa, M. N. Mollah, and E. Hossain, "A comprehensive study of key electric vehicle (EV) components, technologies, challenges, impacts, and future direction of development," *Energies*, vol. 10, no. 8, 2017, doi: 10.3390/en10081217.
- [14] S. Deb, K. Tammi, K. Kalita, and P. Mahanta, "Impact of electric vehicle charging station load on distribution network," *Energies*, vol. 11, no. 1, 2018, doi: 10.3390/en11010178.
- [15] L. I. Dulau and D. Bica, "Effects of electric vehicles on power networks," *Procedia Manufacturing*, vol. 46, pp. 370–377, 2020, doi: 10.1016/j.promfg.2020.03.054.
- [16] A. S. Mogos and S. Grillo, "Impact of EV Charging Stations in Power Grids in Italy and its Mitigation Mechanisms," 21st IEEE International Conference on Environment and Electrical Engineering and 2021 5th IEEE Industrial and Commercial Power System Europe, EEEIC / I and CPS Europe 2021 - Proceedings, 2021, doi: 10.1109/EEEIC/ICPSEurope51590.2021.9584782.
- [17] J. Lingzhi and P. Slowik, "Literature review of electric vehicle consumer awareness and outreach activities," *The international council of clean transportation*, vol. 41, no. November, p. 27, 2017, [Online]. Available: http://0-search.ebscohost.com.biblio.url.edu/login.aspx?direct=true&db=poh&AN=89013221&lang=es&site=ehost-live&scope=site.
- [18] D. R. Pinto *et al.*, "Field investigation of the power quality impact of electric vehicles in secondary residential systems," pp. 1–6, 2018, doi: 10.1109/ichqp.2018.8378904.
- [19] A. G. Anastasiadis, G. P. Kondylis, A. Polyzakis, and G. Vokas, "Effects of increased electric vehicles into a distribution network," *Energy Procedia*, vol. 157, pp. 586–593, 2019, doi: 10.1016/j.egypro.2018.11.223.
- [20] M. P. Jadhav and V. N. Kalkhambkar, "Electric Vehicle for Frequency Regulation of Microgrid," JournalNX, pp. 174–178, 2018.
- [21] G. Durga and S. Painuli, "Effects on Distribution System Voltage Stability including Electric Vehicles and its Enhancement by placing DG at Optimal location," *Biennial International Conference on Power and Energy Systems: Towards Sustainable Energy* (*PESTSE*), p. 6, 2018.
- [22] L. Zhang et al., "Sustainable and high-efficiency recycling of valuable metals from oily honing ferroalloy scrap via de-oiling and smelting separation," *Journal of Hazardous Materials*, vol. 413, 2021, doi: 10.1016/j.jhazmat.2021.125399.
- [23] V. Sandeep, S. Shastri, A. Sardar, and S. R. Salkuti, "Modeling of battery pack sizing for electric vehicles," *International Journal of Power Electronics and Drive Systems*, vol. 11, no. 4, pp. 1987–1994, 2020, doi: 10.11591/ijpeds.v11.i4.pp1987-1994.
 [24] A. Dominic Savio and A. Vimala Juliet, "Development of multiple plug-in electric vehicle mobile charging station using
- [24] A. Dominic Savio and A. Vimala Juliet, "Development of multiple plug-in electric vehicle mobile charging station using bidirectional converter," *International Journal of Power Electronics and Drive Systems*, vol. 11, no. 2, pp. 785–791, 2020, doi: 10.11591/ijpeds.v11.i2.pp785-791.
- [25] Y. Shen, W. Fang, F. Ye, and M. Kadoch, "EV charging behavior analysis using hybrid intelligence for 5G smart grid," *Electronics (Switzerland)*, vol. 9, no. 1, 2020, doi: 10.3390/electronics9010080.
- [26] S. Afshar, P. MacEdo, F. Mohamed, and V. Disfani, "A literature review on mobile charging station technology for electric vehicles," 2020 IEEE Transportation Electrification Conference and Expo, ITEC 2020, pp. 1184–1190, 2020, doi: 10.1109/ITEC48692.2020.9161499.
- [27] S. Vadi, R. Bayindir, A. M. Colak, and E. Hossain, "A review on communication standards and charging topologies of V2G and V2H operation strategies," *Energies*, vol. 12, no. 19, 2019, doi: 10.3390/en12193748.
- [28] P. Dharmakeerthi, "Electric vehicle integration-grid stability concerns and countermeasures," 2014, [Online]. Available: https://espace.library.uq.edu.au/view/UQ:342991/s42472090_phd_submission.pdf?dsi_version=aca2d45b2810b6d8132a79db11e 5bed7.
- [29] S. Deb, K. Kalita, and P. Mahanta, "Review of impact of electric vehicle charging station on the power grid," Proceedings of 2017 IEEE International Conference on Technological Advancements in Power and Energy: Exploring Energy Solutions for an Intelligent Power Grid, TAP Energy 2017, pp. 1–6, 2018, doi: 10.1109/TAPENERGY.2017.8397215.
- [30] M. Zhang, X. Yuan, and J. Hu, "Inertia and Primary Frequency Provisions of PLL-Synchronized VSC HVDC When Attached to Islanded AC System," *IEEE Transactions on Power Systems*, vol. 33, no. 4, pp. 4179–4188, Jul. 2018, doi: 10.1109/TPWRS.2017.2780104.
- [31] A. S. Al-Ogaili *et al.*, "Review on scheduling, clustering, and forecasting strategies for controlling electric vehicle charging: Challenges and recommendations," *IEEE Access*, vol. 7, pp. 128353–128371, 2019, doi: 10.1109/ACCESS.2019.2939595.
- [32] J. A. Peças Lopes, S. A. Polenz, C. L. Moreira, and R. Cherkaoui, "Identification of control and management strategies for LV unbalanced microgrids with plugged-in electric vehicles," *Electric Power Systems Research*, vol. 80, no. 8, pp. 898–906, 2010, doi: 10.1016/j.epsr.2009.12.013.
- [33] Y. Yang, M. El Baghdadi, Y. Lan, Y. Benomar, J. Van Mierlo, and O. Hegazy, "Design methodology, modeling, and comparative study of wireless power transfer systems for electric vehicles," *Energies*, vol. 11, no. 7, 2018, doi: 10.3390/en11071716.
- [34] X. Wu, L. Li, J. Zou, and G. Zhang, "EV-based voltage regulation in line distribution grid," Conference Record IEEE Instrumentation and Measurement Technology Conference, vol. 2016-July, 2016, doi: 10.1109/I2MTC.2016.7520568.
- [35] J. Aghaei et al., "Investigation of smart distribution network response to operation performance of plug-in hybrid electric vehicles," 2017 IEEE PES Innovative Smart Grid Technologies Conference Europe, ISGT-Europe 2017 - Proceedings, vol. 2018-Janua, pp. 1–6, 2017, doi: 10.1109/ISGTEurope.2017.8260304.
- [36] O. Kolawole and I. Al-Anbagi, "Optimizing electric vehicles charging cost for frequency regulation support in a smart grid," 2017 IEEE Electrical Power and Energy Conference, EPEC 2017, vol. 2017-Octob, pp. 1–6, 2018, doi: 10.1109/EPEC.2017.8286190.
- [37] G. Xiao, C. Li, Z. Yu, Y. Cao, and B. Fang, "Review of the impact of electric vehicles participating in frequency regulation on power grid," *Proceedings - 2013 Chinese Automation Congress, CAC 2013*, pp. 75–80, 2013, doi: 10.1109/CAC.2013.6775705.
- [38] S. Amine and O. Mokhiamar, "A study of stability and power consumption of electric vehicles using different modern control strategies," *Alexandria Engineering Journal*, vol. 58, no. 4, pp. 1281–1290, 2019, doi: 10.1016/j.aej.2019.10.010.

- [39] S. Izadkhast, P. Garcia-Gonzalez, and P. Frias, "An aggregate model of plug-in electric vehicles for primary frequency control," in 2016 IEEE Power and Energy Society General Meeting (PESGM), Jul. 2016, no. October, pp. 1–1, doi: 10.1109/PESGM.2016.7741672.
- [40] M. Nour, J. P. Chaves-Ávila, G. Magdy, and Á. Sánchez-Miralles, "Review of positive and negative impacts of electric vehicles charging on electric power systems," *Energies*, vol. 13, no. 18, 2020, doi: 10.3390/en13184675.
- [41] M. Sabarimuthu, N. Senthilnathan, A. M. Monnisha, V. KamaleshKumar, S. Krithika Sree, and P. Mala Sundari, "Measurement and Analysis of Power Quality Issues Due to Electric Vehicle Charger," *IOP Conference Series: Materials Science and Engineering*, vol. 1055, no. 1, p. 012131, 2021, doi: 10.1088/1757-899x/1055/1/012131.
- [42] A. S. Rodrigo and V. G. C. Priyanka, "Impact of high penetration of EV charging on harmonics in distribution networks," *MERCon 2018 - 4th International Multidisciplinary Moratuwa Engineering Research Conference*, pp. 340–344, 2018, doi: 10.1109/MERCon.2018.8421990.
- [43] A. K. Karmaker, S. Roy, and M. R. Ahmed, "Analysis of the Impact of Electric Vehicle Charging Station on Power Quality Issues," in 2019 International Conference on Electrical, Computer and Communication Engineering (ECCE), Feb. 2019, pp. 1–6, doi: 10.1109/ECACE.2019.8679164.
- [44] K. Clement-Nyns, E. Haesen, and J. Driesen, "The impact of Charging plug-in hybrid electric vehicles on a residential distribution grid," *IEEE Transactions on Power Systems*, vol. 25, no. 1, pp. 371–380, 2010, doi: 10.1109/TPWRS.2009.2036481.
- [45] B. Vatandoust, A. Ahmadian, and M. A. Golkar, "Stochastic copula-based multivariate modeling of plug-in hybrid electric vehicles load demand in residential distribution network," 2016 Smart Grids Conference, SGC 2016, pp. 102–108, 2017, doi: 10.1109/SGC.2016.7883460.
- [46] M. S. Elnozahy and M. M. A. Salama, "Studying the feasibility of charging plug-in hybrid electric vehicles using photovoltaic electricity in residential distribution systems," *Electric Power Systems Research*, vol. 110, pp. 133–143, 2014, doi: 10.1016/j.epsr.2014.01.012.
- [47] C. H. Dharmakeerthi, N. Mithulananthan, and T. K. Saha, "Overview of the impacts of plug-in electric vehicles on the power grid," 2011 IEEE PES Innovative Smart Grid Technologies, ISGT Asia 2011 Conference: Smarter Grid for Sustainable and Affordable Energy Future, 2011, doi: 10.1109/ISGT-Asia.2011.6167115.
- [48] E. Sortomme, A. I. Negash, S. S. Venkata, and D. S. Kirschen, "Voltage dependent load models of charging electric vehicles," *IEEE Power and Energy Society General Meeting*, 2013, doi: 10.1109/PESMG.2013.6672752.
- [49] S. Rezaee, E. Farjah, and B. Khorramdel, "Probabilistic analysis of plug-in electric vehicles impact on electrical grid through homes and parking lots," *IEEE Transactions on Sustainable Energy*, vol. 4, no. 4, pp. 1024–1033, 2013, doi: 10.1109/TSTE.2013.2264498.
- [50] P. Papadopoulos, S. Skarvelis-Kazakos, I. Grau, B. Awad, L. M. Cipcigan, and N. Jenkins, "Impact of residential charging of electric vehicles on distribution networks, a probabilistic approach," *Proceedings of the Universities Power Engineering Conference*, 2010.
- [51] P. J. A. Lopes, F. J. Soares, R. P. M. Almeida, P. C. Baptista, C. M. Silva, and T. L. Farias, "Quantification of technical impacts and environmental benefits of electric vehicles integration on electricity grids," 2009 8th International Symposium on Advanced Electromechanical Motion Systems and Electric Drives Joint Symposium, ELECTROMOTION 2009, 2009, doi: 10.1109/ELECTROMOTION.2009.5259139.
- [52] A. Maitra, K. S. Kook, J. Taylor, and A. Giumento, "Grid impacts of plug-in electric vehicles on Hydro Quebec's Distribution system," 2010 IEEE PES Transmission and Distribution Conference and Exposition: Smart Solutions for a Changing World, 2010, doi: 10.1109/TDC.2010.5484352.
- [53] J. Y. Yong, V. K. Ramachandaramurthy, K. M. Tan, and N. Mithulananthan, "Bi-directional electric vehicle fast charging station with novel reactive power compensation for voltage regulation," *International Journal of Electrical Power and Energy Systems*, vol. 64, pp. 300–310, 2015, doi: 10.1016/j.ijepes.2014.07.025.
- [54] F. Geth, N. Leemput, J. Van Roy, J. Buscher, R. Ponnette, and J. Driesen, "Voltage droop charging of electric vehicles in a residential distribution feeder," *IEEE PES Innovative Smart Grid Technologies Conference Europe*, 2012, doi: 10.1109/ISGTEurope.2012.6465692.
- [55] O. Beaude, Y. He, and M. Hennebel, "Introducing decentralized EV charging coordination for the voltage regulation," 2013 4th IEEE/PES Innovative Smart Grid Technologies Europe, ISGT Europe 2013, 2013, doi: 10.1109/ISGTEurope.2013.6695375.
- [56] M. S. Elnozahy and M. M. A. Salama, "A comprehensive study of the impacts of PHEVs on residential distribution networks," *IEEE Transactions on Sustainable Energy*, vol. 5, no. 1, pp. 332–342, 2014, doi: 10.1109/TSTE.2013.2284573.
- [57] J. Susanto, F. Shahnia, and D. Ludwig, "A framework to technically evaluate integration of utility-scale photovoltaic plants to weak power distribution systems," *Applied Energy*, vol. 231, pp. 207–221, 2018, doi: 10.1016/j.apenergy.2018.09.130.
- [58] R. Wei, Z. Liu, K. Zhou, L. Xu, and B. Xu, "Frequency control in distribution feeders based on bidirectional V2G converter for EV," 2017 IEEE Conference on Energy Internet and Energy System Integration, EI2 2017 - Proceedings, vol. 2018-Janua, pp. 1– 5, 2017, doi: 10.1109/EI2.2017.8245242.
- [59] X. Xu, C. Zhang, and L. Gu, "Decentralized primary frequency regulation control strategy for vehicle-To-grid," 2016 3rd International Conference on Systems and Informatics, ICSAI 2016, pp. 217–222, 2017, doi: 10.1109/ICSAI.2016.7810957.
- [60] J. Fang, R. Zhang, H. Li, and Y. Tang, "Frequency Derivative-Based Inertia Enhancement by Grid-Connected Power Converters With a Frequency-Locked-Loop," *IEEE Transactions on Smart Grid*, vol. 10, no. 5, pp. 4918–4927, 2018, doi: 10.1109/TSG.2018.2871085.
- [61] W. Eshetu, P. Sharma, and C. Sharma, "ANFIS based load frequency control in an isolated micro grid," *Proceedings of the IEEE International Conference on Industrial Technology*, vol. 2018-Febru, pp. 1165–1170, 2018, doi: 10.1109/ICIT.2018.8352343.
- [62] A. Rabie, A. Ghanem, S. Kaddah, and M. El-Saadawi, "Frequency Stability in Weak Grids Using Independent Electric Vehicle," in 2019 21st International Middle East Power Systems Conference (MEPCON), Dec. 2019, pp. 423–428, doi: 10.1109/MEPCON47431.2019.9007917.
- [63] H. Liu, J. Qi, J. Wang, P. Li, C. Li, and H. Wei, "EV Dispatch Control for Supplementary Frequency Regulation Considering the Expectation of EV Owners," *IEEE Transactions on Smart Grid*, vol. 9, no. 4, pp. 3763–3772, 2018, doi: 10.1109/TSG.2016.2641481.
- [64] U. Ur Rehman and M. Riaz, "Real time controlling algorithm for vehicle to grid system under price uncertainties," *Proceedings* -2018, IEEE 1st International Conference on Power, Energy and Smart Grid, ICPESG 2018, pp. 1–7, 2018, doi: 10.1109/ICPESG.2018.8384522.
- [65] S. Hashemi, N. B. Arias, P. Bach Andersen, B. Christensen, and C. Traholt, "Frequency Regulation Provision Using Cross-Brand Bidirectional V2G-Enabled Electric Vehicles," 2018 6th IEEE International Conference on Smart Energy Grid Engineering, SEGE 2018, pp. 249–254, 2018, doi: 10.1109/SEGE.2018.8499485.

- [66] H. Liu, Z. Hu, Y. Song, J. Wang, and X. Xie, "Vehicle-to-Grid Control for Supplementary Frequency Regulation Considering Charging Demands," *IEEE Transactions on Power Systems*, vol. 30, no. 6, pp. 3110–3119, 2015, doi: 10.1109/TPWRS.2014.2382979.
- [67] Y. Ota, H. Taniguchi, T. Nakajima, K. M. Liyanage, J. Baba, and A. Yokoyama, "Autonomous Distributed V2G (Vehicle-to-Grid) Satisfying Scheduled Charging," *IEEE Transactions on Smart Grid*, vol. 3, no. 1, pp. 559–564, Mar. 2012, doi: 10.1109/TSG.2011.2167993.
- [68] H. Liu, Z. Hu, Y. Song, and J. Lin, "Decentralized vehicle-to-grid control for primary frequency regulation considering charging demands," *IEEE Transactions on Power Systems*, vol. 28, no. 3, pp. 3480–3489, 2013, doi: 10.1109/TPWRS.2013.2252029.
- [69] Y. Ota, H. Taniguchi, T. Nakajima, K. M. Liyanage, J. Baba, and A. Yokoyama, "Autonomous distributed V2G (vehicle-to-grid) considering charging request and battery condition," in 2010 IEEE PES Innovative Smart Grid Technologies Conference Europe (ISGT Europe), Oct. 2010, vol. 3, no. 1, pp. 1–6, doi: 10.1109/ISGTEUROPE.2010.5638913.
- [70] M. Marinelli, S. Martinenas, K. Knezović, and P. B. Andersen, "Validating a centralized approach to primary frequency control with series-produced electric vehicles," *Journal of Energy Storage*, vol. 7, pp. 63–73, 2016, doi: 10.1016/j.est.2016.05.008.
- [71] P. M. Rocha Almeida, J. A. Peças Lopes, F. J. Soares, and M. H. Vasconcelos, "Automatic generation control operation with electric vehicles," 2010 IREP Symposium - Bulk Power System Dynamics and Control - VIII, IREP2010, 2010, doi: 10.1109/IREP.2010.5563295.
- [72] W. Tian, J. He, L. Niu, W. Zhang, X. Wang, and Z. Bo, "Simulation of vehicle-to-grid (V2G) on power system frequency control," 2012 IEEE Innovative Smart Grid Technologies - Asia, ISGT Asia 2012, 2012, doi: 10.1109/ISGT-Asia.2012.6303105.
- [73] H. Jia et al., "Coordinated control for EV aggregators and power plants in frequency regulation considering time-varying delays," Applied Energy, vol. 210, pp. 1363–1376, 2018, doi: 10.1016/j.apenergy.2017.05.174.
- [74] J. Meng, Y. Mu, H. Jia, J. Wu, X. Yu, and B. Qu, "Dynamic frequency response from electric vehicles considering travelling behavior in the Great Britain power system," *Applied Energy*, vol. 162, pp. 966–979, 2016, doi: 10.1016/j.apenergy.2015.10.159.
- [75] L. Wang and B. Chen, "Dual-level consensus-based frequency regulation using vehicle-to-grid service," *Electric Power Systems Research*, vol. 167, pp. 261–276, 2019, doi: 10.1016/j.epsr.2018.10.022.
- [76] P. M. R. Almeida, F. J. Soares, and J. A. P. Lopes, "Electric vehicles contribution for frequency control with inertial emulation," *Electric Power Systems Research*, vol. 127, pp. 141–150, 2015, doi: 10.1016/j.epsr.2015.05.026.
- [77] J. Donadee and M. Ilić, "Stochastic co-optimization of charging and frequency regulation by electric vehicles," 2012 North American Power Symposium, NAPS 2012, 2012, doi: 10.1109/NAPS.2012.6336373.
- [78] O. Ouramdane, E. Elbouchikhi, Y. Amirat, and E. S. Gooya, "Optimal sizing and energy management of microgrids with Vehicleto-Grid technology: A critical review and future trends," *Energies*, vol. 14, no. 14, 2021, doi: 10.3390/en14144166.
- [79] Q. Guo, J. Han, M. Yoon, and G. Jang, "A study of economic dispatch with emission constraint in smart grid including wind turbines and electric vehicles," 2012 IEEE Vehicle Power and Propulsion Conference, VPPC 2012, pp. 1002–1005, 2012, doi: 10.1109/VPPC.2012.6422753.
- [80] A. Y. S. Lam, K. C. Leung, and V. O. K. Li, "Capacity management of vehicle-to-grid system for power regulation services," 2012 IEEE 3rd International Conference on Smart Grid Communications, SmartGridComm 2012, pp. 442–447, 2012, doi: 10.1109/SmartGridComm.2012.6486024.
- [81] S. Han, S. Han, and K. Sezaki, "Development of an optimal vehicle-to-grid aggregator for frequency regulation," IEEE Transactions on Smart Grid, vol. 1, no. 1, pp. 65–72, 2010, doi: 10.1109/TSG.2010.2045163.
- [82] S. W. Kim, Y. G. Jin, Y. H. Song, and Y. T. Yoon, "Decentralized vehicle-to-grid design for frequency regulation within pricebased operation," *Journal of Electrical Engineering and Technology*, vol. 10, no. 3, pp. 1335–1341, 2015, doi: 10.5370/JEET.2015.10.3.1335.
- [83] S. Martinenas, K. Knezovic, and M. Marinelli, "Management of Power Quality Issues in Low Voltage Networks Using Electric Vehicles: Experimental Validation," *IEEE Transactions on Power Delivery*, vol. 32, no. 2, pp. 971–979, 2017, doi: 10.1109/TPWRD.2016.2614582.
- [84] C. H. Dharmakeerthi and N. Mithulananthan, "PEV load and its impact on static voltage stability," *Power Systems*, vol. 91, pp. 221–248, 2015, doi: 10.1007/978-981-287-299-9_8.
- [85] C. H. Dharmakeerthi, N. Mithulananthan, and T. K. Saha, "Modeling and planning of EV fast charging station in power grid," *IEEE Power and Energy Society General Meeting*, 2012, doi: 10.1109/PESGM.2012.6345008.
- [86] D. Meyer and J. Wang, "Integrating ultra-fast charging stations within the power grids of smart cities: A review," *IET Smart Grid*, vol. 1, no. 1, pp. 3–10, 2018, doi: 10.1049/iet-stg.2018.0006.
- [87] C. H. Dharmakeerthi, N. Mithulananthan, and T. K. Saha, "A comprehensive planning framework for electric vehicle charging infrastructure deployment in the power grid with enhanced voltage stability," *International Transactions on Electrical Energy Systems*, vol. 25, no. 6, pp. 1022–1040, 2015, doi: 10.1002/etep.1886.
- [88] C. H. Dharmakeerthi and M. Nadarajah, "Stability Cogitated Electric Vehicle Charging Infrastructure Planning," *International Journal of Smart Grid and Sustainable Energy Technologies*, vol. 1, no. 1, pp. 10–14, 2019, doi: 10.36040/ijsgset.v1i1.180.
- [89] S. A. Mohamed, "Design and Control of V2G to Enhance System Voltage Stability," Universal Journal of Electrical and Electronic Engineering, vol. 7, no. 1, pp. 1–18, 2020, doi: 10.13189/ujeee.2020.070101.
- [90] M. Nour, A. Ali, and C. Farkas, "Evaluation of Electric Vehicles Charging Impacts on A Real Low Voltage Grid," International Journal on Power Engineering and Energy (IJPEE), no. 9, 2018, doi: 10.12986/IJPEE.2018.006.
- [91] G. Razeghi, L. Zhang, T. Brown, and S. Samuelsen, "Impacts of plug-in hybrid electric vehicles on a residential transformer using stochastic and empirical analysis," *Journal of Power Sources*, vol. 252, pp. 277–285, 2014, doi: 10.1016/j.jpowsour.2013.11.089.
- [92] S. Hajforoosh, M. A. S. Masoum, and S. M. Islam, "Online optimal variable charge-rate coordination of plug-in electric vehicles to maximize customer satisfaction and improve grid performance," *Electric Power Systems Research*, vol. 141, pp. 407–420, 2016, doi: 10.1016/j.epsr.2016.08.017.
- [93] U. Datta, A. Kalam, and J. Shi, "The Strategies of EV Charge/Discharge Management in Smart Grid Vehicle-to-Everything (V2X) Communication Networks," Advanced Communication and Control Methods for Future Smartgrids, 2019, doi: 10.5772/intechopen.85385.
- [94] M. Nour, S. M. Said, H. Ramadan, A. Ali, and C. Farkas, "Control of Electric Vehicles Charging Without Communication Infrastructure," 2018 20th International Middle East Power Systems Conference, MEPCON 2018 - Proceedings, pp. 773–778, 2019, doi: 10.1109/MEPCON.2018.8635277.
- [95] F. Marra et al., "Improvement of local voltage in feeders with photovoltaic using electric vehicles," IEEE Transactions on Power Systems, vol. 28, no. 3, pp. 3515–3516, 2013, doi: 10.1109/TPWRS.2013.2248959.
- [96] M. C. Kisacikoglu, B. Özpineci, and L. M. Tolbert, "Examination of a PHEV bidirectional charger system for V2G reactive power compensation," Conference Proceedings - IEEE Applied Power Electronics Conference and Exposition - APEC, pp. 458–

465, 2010, doi: 10.1109/APEC.2010.5433629.

[97] M. A. Azzouz, M. F. Shaaban, and E. F. El-Saadany, "Real-Time Optimal Voltage Regulation for Distribution Networks Incorporating High Penetration of PEVs," *IEEE Transactions on Power Systems*, vol. 30, no. 6, pp. 3234–3245, 2015, doi: 10.1109/TPWRS.2014.2385834.

BIOGRAPHIES OF AUTHORS



Afaf Rabie D 🔀 🖾 🌣 received the B.Sc. and M.Sc. degrees from Mansoura University, Mansoura, Egypt, in 2006 and 2012, respectively, all in electrical engineering. She is currently pursuing the Ph.D. degree in electrical engineering with the Mansoura University, Egypt. She currently works with North Delta Electricity Distribution Co., Mansoura, Egypt. Her research interests include Stability analysis in weak grids, grid connected electric vehicles and distributed optimization. She can be contacted at email: afaf_fathy2006@yahoo.com.



Abdelhady Ghanem b S received the B.Sc., M.Sc., and Ph.D. degrees from Mansoura University, Mansoura, Egypt, in 2006, 2011, and 2017, respectively, all in electrical engineering. Since 2006, he has been with the Electrical Engineering Department, Mansoura University, where he is now an Associate Professor. He joined the University of Nottingham, Nottingham, U.K., as an Occasional Ph.D. Student (Joint Supervision Scheme between Mansoura University and the University of Nottingham), from February 2015 to May 2017. His research interests include electrical systems modeling, renewable power generation, power system analysis and control, grid-connected power electronics converters, fault diagnosis, and stability analysis. He can be contacted at email: aghanem_m@mans.edu.eg.



Sahar S. Kaddah **(b)** S **(c)** received her B.Sc. (1988) and M.Sc. (1992) from Mansoura University, Egypt. She received her Ph.D. from Howard University, Washington, DC, USA, in 2002. She has been a professor at the Electrical Engineering Department, Mansoura University since 2017. Her research interests are the analysis and control of power systems, renewable energy applications, DG, microgrids, and power system optimization. She can be contacted at email: She can be contacted at email: Email: skaddah@mans.edu.eg.



Magdi M. El-Saadawi b X c was born in Mansoura, Egypt, in 1959. He received the B.Sc. and M.Sc. degrees from Mansoura University, Egypt, in 1982 and 1988, respectively, and the Ph.D. degree from the Warsaw University of Technology, in 1997. He was a Teaching Assistant with El-Mansoura University from 1983–1992. From 1997, he was a Staff Member with the Electrical Engineering Department, Mansoura University, and has been a Professor since May 2011. He was the Head of the Electrical Engineering Department, Mansoura University from 2011 to 2012. His research interests include renewable energy, distributed generation, and power system analysis. He can be contacted at email: m_saadawi@mans.edu.eg.